Author’s response to reviews

Title: "The sun keeps rising but darkness surrounds us": a qualitative exploration of the lived experiences of women with obstetric fistula in Ethiopia

Authors:

shifera yedenekal (shiferaa@yahoo.com)

Misganaw Animut (misganawanimut@gmail.com)

Abebe Gebretsadik (abema24@yahoo.com)

Lakew Gebretsadik (lakewgebretsadika9@gmail.com)

Million Berhe (megaabera@yahoo.com)

Zewdie Koricha (zbkoricha@yahoo.com)

Version: 8 Date: 30 Jan 2019

Author’s response to reviews:

Response to Editor

1. The editor Requested us for copy editing.

Response: - Thank you.

The current manuscript is copy edited by a professional and fluent English language speaker twice. For further inquiry please contact the person who did the copy editing by the following mailing address.

Name: - Nicole Bergen

Email: nberg094@uottawa.ca or nicolebergen@gmail.com, University of Ottawa, 600 Peter Morand Crescent, Ottawa, ON K1G 5Z3, Canada

2. Consent to publish; In your “consent to publish” section, please confirm whether participants gave consent for direct quotes from their interviews to be published in this manuscript.

Response: Thank you.

We included the information as per your request.
3. Availability of data and materials

Response: - Thank you again.
We did the revision accordingly.

4. To upload Clean manuscript

Response: - Dear Editor

Thank you for your comments and suggestions. At this time, we uploaded the cleaned manuscript on your system.

Response to Reviewer

1. The reviewer recommends the manuscript to be copy edited.

Response: - We did the copy editing by professional English speaker.

Finally, on behalf of all authors; I want to pass my acknowledgment to the Editor and reviewers.